Reviewer #1

: 1. Thank you for your sincere advice.

2. After about a week of the ischemic stroke, the hyperacute stage passed and it was judged to be stable, and DCE-MRI was performed one week later. Since this study is a case study, DCE-MRI could not be checked with a perfect protocol. In the next study, we will supplement this part. Thank you very much.

Reviewer #2

: 1. Thank you for your sincere advice.

2. I totally agree with you that a case report cannot draw conclusions. We will conduct better research with more cases.

Science editor

: 1. Thank you for your sincere advice.

2. We corrected the sentence to the introduction.

3. The treatment process has been supplemented a little more, and in the next study, the mechanism for the permeability of blood-brain barrier will be further supplemented.

Company editor-in-chief

1. Thank you for your sincere advice.

2. We prepare and arrange the figures using PowerPoint. Thank you so much.